Author's response to reviews

Title: The use of CAM and conventional treatment among primary care consulters with chronic musculoskeletal pain

Authors:

Majid Artus (m.artus@cphc.keele.ac.uk)
Peter Croft (p.r.croft@cphc.keele.ac.uk)
Martyn Lewis (a.m.lewis@cphc.keele.ac.uk)

Version: 4 Date: 2 May 2007

Author's response to reviews:

Manuscript revised to address the formatting changes suggested by The BioMed Central Editorial Production Team.